**Author's response to reviews**

**Title:** Communicating effectiveness of intervention for chronic diseases: what single format can replace comprehensive information?

**Authors:**

Henrik Støvring (hstovring@health.sdu.dk)
Dorte Gyrd-Hansen (dgh@dsi.dk)
Ivar S Kristiansen (ivarsk@c2i.net)
Jørgen Nexoe (jnexoe@health.sdu.dk)
Jesper B Nielsen (jbnielsen@health.sdu.dk)

**Version:** 5  **Date:** 21 May 2008

**Author's response to reviews:**

Dear Editor,

We hereby submit what we hope to be the final version, which is nearly identical to the previous version that was conditionally accepted for publication. Apart from correcting minor typo's, the only changes are: the abstract has been modified to follow the standard format of the journal, citations have been updated to reflect current publication status, and Tables have been included in the main document. We hope the paper can be unconditionally accepted in its current form.

Kind regards,

Henrik Støvring